Author’s response to reviews

Title: Mental health and psychosocial support services in primary health care in Nepal: Perceived facilitating factors, barriers and strategies for improvement

Authors:

Nawaraj Upadhaya (upadhayan@gmail.com)
Upasana Regmi (regmi.upasana@gmail.com)
Dristy Gurung (dgurung@tponepal.org.np)
Nagendra Luitel (luitelnp@gmail.com)
Inge Petersen (peterseni@ukzn.ac.za)
Mark Jordans (mark.jordans@kcl.ac.uk)
Ivan Komproe (ivan.komproe@health-works.org)

Version: 3 Date: 27 Jan 2020

Author’s response to reviews:

1. Please provide a Keywords section after your Abstract that includes three to ten keywords representing the main content of the article.

Response: The following key words have been included after the abstract. "Mental health and psychosocial support, primary health care workers, Nepal, facilitating factors, barriers".